Dear editor,

Thanks for all your help and kindness. We are appreciated and grateful all the help you have given us from the bottom of our hearts.

Please find the revised version of our manuscript entitled “Managing Immune Checkpoint Inhibitor-Associated Gastritis: Insights and Strategies” by Lili Yu, Zhilin He and Xinlai Qian for your consideration to be published in World Journal of Gastroenterology.

We are very grateful to your evaluation on our manuscript and an opportunity for the revision. We also greatly appreciate the reviewer’s efforts into reviewing our manuscript and kind comments to assist us in preparing the revision. We have revised manuscript based on the editor and reviewer’s comments and suggestions. The changes that we have made were highlighted with yellow color in the revised manuscript.

In this revised manuscript, we reviewed the English and grammar, marking all changes clearly. We include a graphical abstract. We also include the Highlights in a separate file in the online submission system.

Hopefully, we adequately address the reviewer's concerns in the revision. Your and the experts' re-examination and favorable consideration would be most appreciated.

Sincerely yours.

Lili Yu, Ph. D
The following is a point-by-point response to each comment of the reviewer. The page and line numbers cited in the response refer to the revised version of manuscript.

We are very grateful to the reviewer’s efforts into reviewing our manuscript. We greatly appreciate the reviewer’s favorable consideration on our study. We greatly appreciate the reviewer’s positive general comments and the reviewer’s criticisms are very helpful for us to revise manuscript. We have made revision based on the reviewer’s concerns.

1. **Authors did not follow the proper reference format. So, please follow the proper format.**

   We agree with the reviewer’s suggestion. Thank you very much for your carefulness and preciseness. For this part, We have modified it to be a proper reference format.

2. **The abbreviation mentioned initially for proper understanding.**

   We agree with the reviewers suggestion. Thank you very much for your carefulness and preciseness. We have added all the full names to the original abbreviations, and we also added the "abbreviation" section to the article.

3. **The manuscript is not 'spell or grammar checked' and then was unreadable as a scientific paper. The authors should carefully check and brush up their manuscript.**

   We agree with the reviewers suggestion. Thank you very much for your carefulness and preciseness. On this point, we have corrected the spelling mistakes.
Once again, we are very grateful to the reviewer’s and editorial comments for assistances in preparing the revision.